\documentclass[]{beamer}
\usetheme{Rochester} 
\usepackage{graphicx}
\definecolor{links}{HTML}{2A1B81}
\hypersetup{colorlinks,linkcolor=,urlcolor=links}
\usepackage{url}
\usepackage{color}
\usepackage{wrapfig}
\usepackage{listings}
\setbeamertemplate{navigation symbols}{} 
\usepackage{graphicx}
%\usepackage[natbib=true]{biblatex}
%\bibliography{/GW/root/golem,/GW/root/others} % or


\AtBeginSection[]
{
   \begin{frame}
       \frametitle{Outline}
       \tableofcontents[currentsection,currentsubsection]
   \end{frame}
}

%\def\GWp{/home/svoboda/backups/home/svoboda/GOLEM/GW/}
\def\GWp{/home/svoboda/buon/GW/}


\def\slide#1#2#3{\begin{frame}[label=#1]\frametitle{#2} #3\end{frame}}

%\def\GWinput#1{\input{\GWp/#1}}
\def\GWinput#1{\immediate\write18{mkdir -p GW/`dirname #1`; cp  \GWp/#1  GW/#1}\input{GW/#1}}

%\def\GWincludegraphics#1#2{\includegraphics[#1]{\GWp/#2}}
\def\GWincludegraphics#1#2{\IfFileExists{GW/#2}{\immediate\write18{echo "";echo "*************** #2 exists, no wget *******************"}}{\immediate\write18{mkdir -p GW/`dirname #2`;cp \GWp/#2 GW/#2;}}\includegraphics[#1]{GW/#2}}

\def\GWparam#1{\input{\GWp/#1}}
\def\tw{\textwidth}

\def\InOut#1{#1}%Just for tuning
\def\InOutI#1{}%Just for tuning 

\begin{document}
\title{Introduction to tokamak operation \\ (GOLEM specific) - Level 1}
\author{Vojt\v{e}ch Svoboda \\ on behalf of the tokamak GOLEM team}
\InOut{
%\GWinput{Tokamak/Lectures/Commons/latexsrc.page}\LetsCountDown
\GWinput{Production/Lectures/SpecificOpenings/latexsrc.page}\SunPlasma\FusionMission\FusionMilestones\GOLEMStartLine
\slide{title}{Title}{\titlepage}}
\InOut{\section{Introduction}
\slide{}{Basic characteristics}{
  \GWinput{Tokamak/BasicCharacteristics/latexrsrcs.page}
  \GolemBaseCharacteristicsWithPhoto{0.6\tw}{0.4\tw}
}
\GOLEMhistory
  \GWinput{Tokamak/Introduction/latexsrc.page}
  %\GOLEM_RabiLoew
\GWinput{Tokamak/Overview/latexsrc.page}
  \GolemTokamakRealSouthFebXII
  \GolemTokamakRealSouthFebXIIdescr
  \GolemTokamakRealNorthFebXII
  \GolemTokamakRealNorthFebXIIdescr
  \InfraRoom
  \InfraRoomdescr
  \GWinput{Tokamak/VirtualModel/OnPl/latexsrc.page}\VirtOverwiev}
  \InOut{\section{Tokamak GOLEM - engineering scheme}
\GWinput{Infrastructure/Scheme/latexsrc.page}
  \tokamaktodo
  \engineerinschemeGolemEN
  \engineerinschemeGolemENlevelZ
  \engineerinschemeGolemENlevelZa
  \engineerinschemeGolemENlevelZb
  \vsuvkaLCEN
  \engineerinschemeGolemENlevelIa
  \engineerinschemeGolemENlevelI
%  \engineerinschemeGolemENlevelII
%  \engineerinschemeGolemENlevelIII
 \GWinput{Infrastructure/TriggerSystem/latexsrc.page}\triggersequenceB_CD
\GWinput{Infrastructure/Preionization/latexsrc.page}\PreionizationElFilam
 \GWinput{Tokamak/RemoteControl/latexsrc.page}\VirtControlRoomLI}
 
\section{Tokamak GOLEM - diagnostics}
\GWinput{DAS/latexsrc.page}\GolemDASaNI 
\GWinput{Diagnostics/Basic/Uloop/latexsrc.page}\LoopVdescrSlide
\GWinput{Diagnostics/Basic/ToroidalMagneticField/latexsrc.page}\BtordescrSlide
\GWinput{Diagnostics/Basic/PlasmaPlusChamberCurrent/latexsrc.page}\IpchSlide
\GWinput{Diagnostics/Basic/VisibleRadiation/latexsrc.page}\VisiRadSlide
\GWinput{ShowRooms/PlasmaPerformance/TextBookPlasma/latexsrc.page}\TextBookShotNo11722
\InOutI{
\section{Data access} 
\GWinput{Diagnostics/Basic/latexsrc.page}\NIturbo
\GWinput{SW/wget/latexsrc.page}\wgetI
\GWinput{SW/DataAccess/latexsrc.page}\DataAccess\GOLEMdataavailibility
\section{Data manipulation}
\GWinput{SW/DataAccess/latexsrc.page}\Graphsurvey
\GWinput{SW/matlab/latexsrc.page}\matlabI
\GWinput{SW/octave/latexsrc.page}\octaveI
\GWinput{SW/gnuplot/latexsrc.page}\gnuplotI
\GWinput{SW/python/latexsrc.page}\pythonI
\GWinput{SW/mathematica/latexsrc.page}\mathemI
\GWinput{SW/excel/latexsrc.page}\excelI
\GWinput{SW/gnuplot/latexsrc.page}\gnuplotII
\section{Possible tasks}
\GWinput{Education/ExperimentMenu/BreakdownStudies/latexsrc.page}\MenuBreakdownStudies
\GWinput{Education/ExperimentMenu/H2vsHeWorkingGases/latexsrc.page}\MenuIsotopicStudies
\GWinput{Education/ExperimentMenu/PlasmaPositionMonitoring/FourMirnovCoils/latexsrc.page}\MenuPlasmaPositionStudies
\GWinput{Education/ExperimentMenu/BasicTokamakMeasurements/CentralElectronTemperatureSpitzerFormula/latexsrc.page}\MenuCentralElectronTempSpitzerFormula
\GWinput{Education/ExperimentMenu/VesselConditioning/latexsrc.page}\MenuVesselConditioningStudies
%UZJEFAKTPOZDE
\GWinput{Education/ExperimentMenu/private/diagnostics.tex}
\GWinput{Education/ExperimentMenu/private/experiments.tex}
\FastCameras
\HXRLenkaKXIIphoto
\HXRLenkaKXII
\RakeProbeXII
\RakeProbeXIIresults
\SpectrumTOXII
}


\section{Conclusion}

\slide{Production}{Production}{
\begin{itemize}
 \item Everything via \url{http://golem.fjfi.cvut.cz/current}
 \begin{itemize}
 \item This presentation
  \item Control rooms
  \item Contact: Vojtech Svoboda, +420 737673903, 
  \item possible chat: vojtech.svob@gmail.com
 \end{itemize}
 \item Any shot from mobile phone?
\end{itemize}
}


%\GWinput{Tokamak/Lectures/Acknowledgements/latrexsrc.page}\ackXIV
\nocite{FusenEngDes11}
\nocite{EPSDublin2010}
\nocite{BrotancePhD}
\nocite{EPSStrasbourg2011}
\nocite{wwwgolem}

%\begin{frame} \frametitle{References}
\bibliographystyle{plain}
\bibliography{/GW/root/golem,/GW/root/others}
%\end{frame}

%\printbibliography

\end{document}





